Author’s response to reviews

Title: Adhesive Capsulitis of the Shoulder: Protocol for the Adhesive Capsulitis Biomarker (AdCaB) Study

Authors:
Richard Page (richard.page@deakin.edu.au)
Sean McGee (sean.mcgee@deakin.edu.au)
Kevin Eng (kevineng@me.com)
Graeme Brown (gcbrown@me.com)
Sally Beattie (SALLYB@BarwonHealth.org.au)
Fiona Collier (fmcol@deakin.edu.au)
Stephen Dean Gill (stephen.gill2@deakin.edu.au)

Version: 2 Date: 11 Mar 2019

Author’s response to reviews:

Dear Bethan

Thank you for the opportunity to publish with BMC Musculoskeletal Disorders.

We have updated the manuscript according to your instructions. We drafted as summary table of the analysis as requested, but decided not to include this in the submission. Because of our multi-stage analyses, the table needed to be extensive in order to meaningfully communicate the analyses and reached a point of being similar to the text. We felt that the table was duplicative, and offered little efficiency to the reader. We hope that you find this acceptable.

Kind regards

Stephen